Author's response to reviews

Title: Targeting young drinkers online: The effectiveness of a web-based brief alcohol intervention in reducing heavy drinking among college students: Study protocol of a two-arm parallel group randomized controlled trial

Authors:

Carmen V. Voogt (c.voogt@bsi.ru.nl)
Evelien A. P. Poelen (e.poelen@bsi.ru.nl)
Marloes Kleinjan (m.kleinjan@bsi.ru.nl)
Lex A. C. J. Lemmers (llemmers@trimbos.nl)
Rutger C. M. E. Engels (r.engels@bsi.ru.nl)

Version: 2 Date: 8 April 2011

Author's response to reviews:

April 8, 2011

Dear editor,

The following changes are made:

- The contributions of all authors are described.
- Two references that are submitted and in press are now only in the text in the following way:
  o (Boon B, Risselada A, Huiberts A, Smit F: Reduced alcohol consumption in male adults due to a one time computer tailored advice: A randomised controlled trial, in press).
- All the references in the reference list contain up to 30 authors' names for each citation. The term 'et al.' is not used anymore in the following reference:
- The image file of the figure does not include the title and figure number anymore and is cropped as closely as possible.

I hope that the manuscript now meets the journal’s requirements and I am looking forward to hearing from you.

Yours sincerely,
Carmen V. Voogt:

Correspondence may be sent to Carmen V. Voogt at the subjoined address or via e-mail at: c.voogt@bsi.ru.nl.

Behavioural Science Institute
Radboud University Nijmegen
P.O. Box 9104
6500 HE Nijmegen
The Netherlands
Fax: 024 36 12776